**Author's response to reviews**

**Title:** An exploratory pilot study to assess self-perceived changes among social assistance recipients regarding employment prospects after receiving dental treatment

**Authors:**

Sonica Singhal (sonica.singhal@mail.utoronto.ca)
Muhammad Mamdani (MamdaniM@smh.ca)
Andrew Mitchell (andrew.mitchell4@sympatico.ca)
Howard Tenenbaum (Howard.Tenenbaum@dentistry.utoronto.ca)
Carlos Quiñonez (carlos.quinonez@utoronto.ca)

**Version:** 2  
**Date:** 11 May 2015

**Author's response to reviews:**

1.) List of abbreviations provided

Abbreviations are used in the text, they are defined in the text at first use, and a list of abbreviations are provided, preceding the competing interests and authors’ contributions.

2.) Line numbering provided